DRAFT_2
EO-1 Weekly Status Week of March 10–March 16, 2011
Day of Year 069 - 075
Mission Day 3772 - 3778
Earth Observing One (EO-1) - General
Scheduled 110 science Data Collection Events (DCEs) this week.  
INSTRUMENTS

All instruments operated nominally this week 
· Performed instrument decontamination cycles

· Conducted Hyperion deicing on March 14 from 00:25z to 15:15z

· Conducted ALI outgassing on March 14 from 00:35z to 15:25z

· Performed Instrument calibration

· Conducted Solar Array Characterization on March 14 at 09:36:25z
· Conducted Hyperion Solar Calibration on March 15 at 10:15:37z
Command and Data Handling (C&DH)

EO-1 has continued to experience problems playing back engineering data from the Solid-State Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.
On Thursday, March 10, there was an ESTO AIST 2008 SensorWeb 3G collaborators teleconference.  

Participants:  Dan Mandl, Stu Frye, Pat Cappelaere, Rob Sohlberg, Steve Chien, Vuong Ly, Justin Rice, Matt Handy, and Joe Young.

Notes from this teleconference are as follows:

1. Flight of an Ames ER-2 aircraft has been approved for the summer of 2012 to fly the Brandywine hyperspectral instrument called EMASS.  Dan Mandl is proposing to integrate WCPS and SpaceCube with the EMASS instrument in order to flight validate the HyspIRI IPM.
2. Dan Mandl took the following three actions with respect to this flight experiment:  (1)  Determine JPL’s role in determining which algorithms to run on-board and how much support will be required, (2)  Determine the role of Ames Research Center in interfacing APPLANIX attitude data to SpaceCube and how to output data products to real-time downlink, and (3)  Determine how much of existing HyspIRI IPM funding to allocate for WCPS/SpaceCube integration with EMASS versus funding for JPL and Ames support requirements.
3. There were discussed three approaches to parallelize highly time consuming operations (Pan-sharpening for example).  They are the following:

· Leverage Internal Threads
· Leverage Multi-Core CPUs

· Leverage Scalability of Virtual Machines

· Leverage FPGA’s

4. There was a discussion about the Flood Dashboard for Namibia Flood Pilot.  Matt Handy has been implementing the Flood Dashboard on the University of Illinois Cloud facility and restarting daily updates to river gauge hydrographs, MODIS flood overlays, and observations made by EO-1 and RADARSAT-2.  He is also displaying daily outputs from the TRMM based flood forecast model.  
On March 15, the EO-1 Mission Science Office (MSO) conducted a meeting chaired by Betsy Middleton.  The participants were Betsy Middleton, Petya Campbell, Fred Huemmrich, Larry Corp, Lawrence Ong, Nathan Pollack, David Landis, Yen-Ben Cheng, Qingyuan Zhang, Hank Margolis, Dan Mandl, Stu Frye, and Joe Young.  

Topics pertinent to EO-1 Operations that were discussed are as follows:

1. Stu Frye reported that the Cross Directorate Executive Committee eliminated the Nereids mission proposal from consideration for the EV-2 Venture Class call as a GSFC proposal.
2. Betsy Middleton inquired about who had registered for the May HyspIRI Symposium at GSFC and the need to submit paper abstracts very soon.

3. Betsy instructed that the science imaging schedule be switched to the growing season in the Northern Hemisphere and add new sites in the Southern Hemisphere this coming November.  
4. Betsy instructed that the Senior Review team anticipate what questions will be asked by the Senior Review panel and create one slide per question.  Example questions are what will be EO-1’s primary science goals after a precession of the MLT to earlier than 9:30 am, and is the answer more night time emitters or lunar cals?  Betsy will call a meeting to review the slides.
5. Stu Frye and David Landis are to work with USGS to determine why several thousand EO-1 scenes, primarily for early in mission, were included in original data base dump on August 2009 but are not in latest dump and are not available to Earth Explorer or GLOVIS.
6. Lawrence Ong was given action to work with Steve Ungar and NOAA on Hyperion Lunar Cal characterizations.
7. Status of the work on atmospheric correction was discussed.  Stu stated that Vuong Ly has implemented ATREM for Hyperion and FLAASH for both Hyperion and ALI and they are currently available on EO-1 GeoBliki.com but will moved to the University of Illinois cloud facility now that it is operational.  Petya Campbell stated the outcome of comparing the results produced by various atmospheric correction codes.  This comparison shows the relative strengths and weaknesses of each code but does not determine relative accuracies.
8. Betsy requested that Stu Frye contact Elijah Ramsey and request that he give us an evaluation of his atmospheric correction studies.

9. Next MSO meeting will be March 29.

Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

No change from the previous week.
Mission Planning

· Continued testing of ASPEN to replace the legacy MOPSS mission planning software

· Continued functional testing of ASIST SCP to replace legacy CMS mission planning software

· Found a problem with a couple RTSs that still display incorrect times in the report

· Comparing Hobart ftf files created on SCP with Hobart ftf files created on CMS

· Comparing TDRS ftf files created on SCP with TDRS ftf files created on CMS

· Brought over some RTSs from CMS 

· Made XML files for the RTSs brought over from CMS

· Created the back-up loads on SCP, with correct times on the generated XML output report

· Compared the XML report from SCP to the ipr report from CMS, both matched

· Compared the ftf files from SCP to the ftf files from CMS, both matched

· Developer is working on a change that will fix the flipped bits

· Also working with the developer to find be able to change just the xml output file

· As of now it is generated automatically when the RTS is created

Trending
No change from previous week.
Flight Dynamics
· Observing a high convergence value for the processed tracking data

System Administration
· Continued effort to maintain EO-1 Data Processing during absence of Tech Engineer.

· Performed multiple tape creations

· Created 4 vdump level 0 tapes for local storage and backup of data.

· Created 4 DLT tapes formatted for the science community.

· EO-1 scientist received tapes for science processing.

· JPL reported that some scenes/images were not available for March 1-4.  Working with Tech Engineer guidance to re-process the missing scenes on those days.

· Recreating the contacts directories for the missing scenes/passes

· Populating these directories with the necessary files.

· Manually running the scripts to process these images and send the raw data to EDC facility for archival.

· Currently all scenes from DOY 60, 3/7 scenes from March 1, and 1/7 scenes from March 2 have been processed.

· This process is ongoing.

· Archiving the dps user’s mail as a log of the at/cron jobs related to processing of scenes.

· EO-1 MOC Move

· Attended a kickoff meeting with the FOT to discuss the move at a high level and start determining what documentation that will be needed.

· Attended a second meeting with management and hardware personnel. During this meeting high level details were provided for the objectives, approach, and assumptions of the MOC move.

· Continued ClamAV installations 

· Installed, configured, and tested ClamAV on an additional computer.

· Created full TAR backups of the other legacy mission planning computer.

· Created backups for the primary ASIST computers, new flight dynamics computers, ASPEN computers, countdown clock, new data processing computers, and MOC CNE computer.

· Created weekly tape.

GROUND AND SPACE NETWORK

Station Downtimes

· SG-1 has been declared red until further notice.
· MGS red for planned maintenance (test passes beginning).
Operational Discrepancies
· None to report at this time.
UPCOMING EVENTS

· Inclination maneuvers on hold until further notice

· Next lunar calibration on March 20 at 02:54:52z
Imagery Status

Scenes and Engineering Cals planned for week of March 10, 2011-March 16, 2011      110 
Total scenes and engineering calibrations planned for entire mission – approximately 59,043
Total Scenes:  ALI scenes in the Level 0 archive              54,338                (as of March 16, 2011)

                         Hyperion scenes in the Level 0 archive    54,085               
PUBLICATIONS AND PRESENTATIONS STATUS 
430 publications 

295 presentations

53 articles and press releases[image: image1.png]
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